REVIEW

Journal of Eating Disorders



Medical instability in typical and atypical adolescent anorexia nervosa: a systematic review and meta-analysis



Cliona Brennan^{1,2*}, Sarah Illingworth², Erica Cini^{3,4,5} and Dee Bhakta²

Abstract

This review investigates the relationship between weight and risk of medical instability (specifically bradycardia, hypotension, hypothermia, and hypophosphatemia) in adolescents with typical and atypical anorexia nervosa. Atypical anorexia nervosa, listed as an example under the DSM-5 category of Other Specified Feeding and Eating Disorders (OSFED), describes patients who are not clinically underweight but otherwise meet criteria for anorexia nervosa. There is a lack of empirical evidence exploring medical complications in adolescents presenting with atypical anorexia nervosa. The small number of studies that do exist in this area indicate that medical instability exists across a range of weights, with weight loss being associated with increased medical risk, independent of underweight. The aim of this review was to collate and analyse results from available studies and identify indicators of medical risk in these two groups of adolescents with restrictive eating disorders. Studies were identified by systematic electronic search of medical databases, including PubMed and EMBASE. All studies investigated the relationship between weight and medical instability and included adolescents diagnosed with anorexia nervosa or atypical anorexia nervosa. One randomised controlled trial, five cohort studies and three chart reviews were included, with a total sample size of 2331 participants. Between 29 and 42% of participants presented with medical instability requiring hospitalisation, in the absence of underweight. Underweight adolescents were significantly more likely to have lower blood pressures (p < 0.0001) and bradycardia was significantly associated with greater weight loss (p < 0.05). There were no statistically significant associations found between degree of underweight and heart rate, temperature, or rate of weight loss (p = 0.31, p = 0.46 and p = 0.16, respectively). Adolescents that were less than 70% median body mass index were significantly more likely to have hypophosphatemia (p < 0.05). The findings of this review support the hypothesis that medical instability can occur across a range of weights in adolescent eating disorders, with rapid weight loss being an important indicator of increasing medical risk. Results were limited by the small number of existing studies that contained data for statistical analysis. Rapid weight loss should be considered as an important indicator of medical instability in adolescents presenting with both typical and atypical anorexia nervosa.

Keywords Anorexia nervosa, Atypical anorexia nervosa, Adolescent eating disorders

Plain English Summary

This review compares physical health markers of underweight and healthy weight in young people with restrictive eating disorders. Previous studies on this topic have found that young people with an eating disorder may become

*Correspondence: Cliona Brennan cliona.brennan@slam.nhs.uk Full list of author information is available at the end of the article



© The Author(s) 2023. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

physically unwell regardless of their body weight. The aim of this review was to bring together the results from these studies to establish if the presence of underweight is necessary to become physically unwell. Studies within this review included a total of 2331 adolescent male and females diagnosed with an eating disorder. Roughly one third of young people with an eating disorder were found to be physically unwell and admitted to hospital but were not underweight, and this was similar to the amount of young people that were underweight and needed a hospital admission. Although some markers of physical health were worse in the underweight group, adolescents who were less underweight were also likely to become physically unwell. Overall, the review found that young people can

become physically unwell regardless of whether they are clinically underweight or not, but that more research is needed on this topic.

Background

Advancements in research in eating disorders have highlighted the importance of early identification of the illness and management of associated risks [1-3]. Degree of underweight as a definitive factor of illness severity has been challenged [4], whilst rapid weight loss, in the absence of underweight, has been identified as a key factor leading to medical instability and hospital admission in this patient group [5]. Further research in the area of weight and illness severity has been recommended by authors in the field, to improve assessment and management of atypical eating disorders [6]. The aim of this review was to explore the relationship between weight (% median BMI, absolute weight, speed and magnitude of weight loss) and specific markers of medical instability in adolescents presenting with typical and atypical anorexia nervosa.

Introduction

Diagnosis of restrictive eating disorders

Historically, in the 4th edition of the DSM, AN diagnosis was limited to those below a BMI of 17.5 kg/m^2 in adults. In children and adolescents, percentage median BMI and percentage IBW are two widely used expressions of degree of underweight in adolescent eating disorders research [7]. However, there is currently no consensus on the best method to calculate IBW and different methods have been reported through the literature [8]. AN

diagnosis in those under 18 years of age, was limited to those considered to be underweight, with a median BMI of < 85% being listed as an example of this in the DSM-IV [9]. This led to an assumption that specific weights were necessary to meet full criteria for illness. DSM-5 removed such examples and supported the use of clinical judgement in view of the overall presentation in diagnoses.

Amenorrhea was included within these diagnostic criteria [9]. These rigid criteria resulted in cases frequently being classed as subthreshold for diagnosis, despite presenting with clear symptoms requiring treatment, and were then categorised as EDNOS (Eating Disorders Not Otherwise Specified), as discussed further below. EDNOS was most frequently diagnosed in children and adolescents, as they typically presented with symptoms that did not meet full criteria for AN [10]. EDNOS encompasses a broad range of subthreshold disorders, and includes restrictive, binge-purge and non-underweight ED presentations. This heterogeneous diagnosis was problematic, with prognosis unknown, and treatment needs being unclear, thus unable to be targeted effectively [11].

In the updated DSM-5, diagnostic criteria for AN assessed severity based on both body mass and weight loss, rather than degree of underweight alone, and in the absence (or presence) of amenorrhea [12]. The presence of underweight is now based on clinical judgement, in the context of the overall presentation [12]. Nearly one third of patients formerly diagnosed with EDNOS are estimated to now meet criteria for AN according to the DSM-5 [13]. Improved specificity in classification aids earlier identification and improved management of the illness [1].

Table 1 DSM-5 Diagnostic criteria of anorexia nervosa and OSFED—Example 1. Atypical anorexia nervosa [12]

5	
Anorexia nervosa	OSFED -Example 1. Atypical anorexia nervosa
A Restriction of energy intake relative to requirements leading to a signifi- cantly low body weight in the context of age, sex, developmental trajectory, and physical health B Intense fear of gaining weight or becoming fat, even though underweight C Disturbance in the way in which one's body weight or shape is expe- rienced, undue influence of body weight or shape on self-evaluation, or denial of the seriousness of the current low body weight	

Patients who are not clinically underweight, but have lost significant amounts of weight, and otherwise resemble all characteristics of AN are now diagnosed with Other Specified Feeding or Eating Disorder (OSFED) (Table 1) [12, 14]. Atypical anorexia nervosa (AAN), a term used clinically to describe this patient group, is listed as an example of a presentation within OSFED. AAN remains a relatively new presentation, with a limited but quickly expanding evidence base related to its management and outcomes [4, 5, 15]. For the purpose of this review, the term AAN will be used to describe this group of non-underweight patients presenting with significant weight loss.

Medical complications in AN

AN has, historically, had the highest mortality rate amongst all psychiatric conditions [16], with most deaths occurring between the ages of 16–29 years [17]. It is the third most common chronic illness among adolescent females [17, 18]. Restrictive eating, malnutrition and chronic low weight, all characteristic of AN, result in disturbances of the brain, reproductive, cardiovascular, gastrointestinal and skeletal systems [18–20].

Risk factors for serious illness or death are well defined in adolescent AN [21]. Key risk parameters are identified in the MEED national eating disorders guidance report (Table 2), alongside guidance on the recommended management. Medical consequences of AN in adolescence are described across many studies [1, 18, 22]. Significant bradycardia and low blood pressure, in particular, are common in AN, with these complications reported in both community and acute settings [23].

Bradycardia is defined by a heart rate below 60 bpm, and is considered as significant bradycardia once heart rate falls below 50 bpm [24]. This complication has been shown to quickly resolve with nutritional rehabilitation and weight gain [25]. Degree of underweight and speed and magnitude of weight loss precede bradycardia, with research supporting weight loss, even in the absence of underweight, as a significant predictor [5]. Assalone and colleagues concluded in their study in 2021, that 'recent weight loss is probably the most important determinant of severe bradycardia in adolescents with AN' [26]. Similarly, cardiovascular complications such as hypotension and ECG abnormalities are widely reported in the literature in adolescent AN, and across a range of body weights [4, 27].

Hypophosphatemia, a hallmark feature of the refeeding syndrome, can occur during nutritional rehabilitation in adolescent AN [28]. Although prevalence of the syndrome across all ages in patients with AN is low (6–22%), it is associated with high morbidity and mortality [29]. Highest physical risk has been attributed to adolescents below 70% median BMI, presenting with rapid weight loss and severe energy restriction [30–32].

Medical complications in AAN

Research and evidence relating to medical complications of AAN is limited [33]. Available literature relating to this subset of patients indicates that physical and psychological morbidity is similar to AN [15]. Adolescents with AAN frequently require acute hospital admission to treat medical complications related to weight loss and malnutrition, despite the absence of severe underweight [5].

A recent review, by Freizinger and colleagues, exploring assessment and treatment in adolescent AAN, in comparison to AN concluded that malnutrition related to rapid or significant weight loss, in the absence of underweight, poses similar risks to health as typical presentations of AN [34]. Similarly, a recent Delphi study exploring research priorities in AAN, identified medical complications arising from the disorder as a high priority [35].

Current review

There is a lack of empirical evidence exploring medical complications in atypical anorexia nervosa presentations. The small number of studies and one existing systematic review in this area indicate that medical instability is

 Table 2
 Summary of MEED risk assessment framework of medical parameters [24]

Risk Domain	Parameter	High risk	High concern	Moderate risk
Body mass	%mBMI	< 70%	70–80%	>80%
	Weight loss	1 kg/week	500–999 g/week	< 500 g/week
Cardiovascular health	Heart rate	<40 bpm	40–50 bpm	>50 bpm
	Blood Pressure	Standing systolic BP below 0.4th centile for age	Standing systolic BP below 0.4th centile for age	Normal standing systolic BP for age
	ECG	>460 female,>450 male	>460 female, >450 male	<460 female, <450 male
Hypothermia	Temperature	< 35.5	< 36	>36
Biochemical abnormalities	Phosphate	Low	Low	Normal

similar across underweight and non-underweight adolescents [36]. This review draws together the evidence relating to weight parameters and risk of medical instability in typical and atypical anorexia nervosa.

Methods

Search strategy

Retrospective data from published studies related to weight and physical risk in young people with typical and atypical anorexia nervosa was systematically reviewed. Relevant studies were identified through electronic searches of databases including PubMed, MED-LINE and EMBASE, completed in September 2022, using key search words (anorexia nervosa, atypical anorexia nervosa, medical instability, bradycardia, hypotension, hypothermia, and hypophosphatemia). The process of selection and inclusion of studies in this review is displayed in Fig. 1. Two reviewers worked independently on study selection and agreed upon studies that met inclusion criteria for the review.

Studies were included if they involved research on children and adolescents, between 10 and 18 years old, presenting with AN or AAN, and investigated outcomes related to weight and acute markers of medical instability. Markers of weight included degree of underweight (expressed as %mBMI and %ideal body weight), speed and magnitude of weight loss (in kilograms). Medical instability outcomes recorded were heart rate, blood pressure, temperature, rate of weight loss and serum phosphorous. Studies were excluded if they were not in the English language, did not involve young people with AN, were not in peer reviewed journals or did not include markers of weight and outcomes relating to medical instability. References of included articles were screened for eligibility and included in the review if appropriate.

Two reviewers worked independently to extract data from studies selected for inclusion in the review.



Fig. 1 Selection of studies for this systematic review of the literature

Corresponding authors of studies with missing data required for the meta-analysis were contacted. Studies were excluded from statistical analysis where missing data were not supplied; results of these studies were included in the narrative review. Data were recorded on a Microsoft Excel spreadsheet. Data extracted from studies included first authors name, year of publication, study design, sample size (including breakdown by diagnosis), each weight marker (degree of underweight and rate and magnitude of weight loss, and each outcome relating to medical instability (i.e., HR, BP, temperature, rate of weight loss and serum phosphorous).

Meta-analysis was carried out to compare medical instability outcomes between adolescents with AN and AAN. For this reason, only studies comparing two independent groups that contained data on medical instability outcomes could be included for statistical analysis. Groups were classified by their body mass; the AN group were classified by mBMI < 85%, and the AAN group as having mBMI > 85%.

Data were analysed using Review Manager version 5.4. The Mantel–Haenszel statistical model was used to compare means and standard deviations of the outcome measures (weight loss, HR, temperature, and BP, serum phosphorous) between the AN and AAN groups. Heterogeneity of the data were measured by chi-square. Significant heterogeneity was determined by a p value of less than 0.1 for the chi squared test for heterogeneity [37]. A random effects model was used for the full data set to account for significant heterogeneity. Sensitivity and specificity study outcomes were plotted onto a forest plot with the 95% confidence interval (CI). Statistical significance was determined by a p value of less than 0.05.

Results

Search results

An initial search of databases yielded 292 studies. Following the removal of 114 duplicates, the titles, and abstracts of the remaining 178 articles were screened. Ninetyfour articles were identified for full paper review and were assessed for eligibility, of these a further 85 articles were excluded. Reasons for exclusion were a lack of relevant outcomes on medical instability (n=53), subjects were not children or adolescents (n=19), study did not include weight [11], and study was not specific to AN or AAN (n=2).

The characteristics of the nine included articles are displayed in Table 3. All nine articles were based on adolescent subjects presenting with AN and AAN. All articles included outcomes related to weight and medical instability (HR, BP, temperature, and serum phosphorous). Three articles were included in statistical analysis, the remaining six articles were excluded as they did not include defined groups whose outcomes could be statistically compared.

All nine studies investigated the relationship between weight and medical instability in adolescent anorexia. Three main themes were identified across the articles: degree of underweight and medical instability, weight loss and medical instability, and degree of underweight and hypophosphatemia (in the context of refeeding syndrome risk).

All studies included adolescent participants, and these were mostly female. All nine studies included participants with AN. Four of the nine studies included patients with AN and AAN (diagnosed with EDNOS or OSFED), and two of the nine studies included patients with AN, EDNOS/OSFED and BN. Sample sizes ranged from 29 participants to 1310. Study designs included one randomised controlled trial, five cohort studies and three chart reviews.

Degree of underweight and medical instability

Four studies focused on the relationship between degree of underweight and medical instability. The observational study by Hudson et al. [1] included participants diagnosed with AN, BN, EDNOS or OSFED (n=208). Parameters of medical instability were available for 89% (n=185) of all cases. Thirty five percent of these cases (n=65) had medical instability upon presentation to hospital. Sixty percent had bradycardia (n=39), 54% had hypotension (n=35), 34% had dehydration (n=22), and 26% had hypothermia (n=17). Over half of cases (51%, n=104) required hospital admission at diagnosis. Nearly half of cases (42%, n=75) were medically unstable but were not underweight. Anthropological indices alone were deemed to be unreliable markers of medical stability in adolescents with EDs.

Sawyer et al. (2016) and Whitelaw et al. (2014) similarly found that markers of medical instability were not limited to underweight adolescents, with higher weight adolescents also frequently requiring hospital admission. In the study by Sawyer et al. (2016), over 40% of adolescents presenting with AAN required hospitalisation. Bradycardia, hypotension, and hypothermia rates were not significantly different between adolescent with AN or AAN. Adolescents with AAN lost more weight than those with AN (17.6 kg vs. 11.0 kg) over a longer period (13.3 vs. 10.2 months). The AAN group had a greater rate of weight loss than those with AN (1.3 kg/month vs. 1.1 kg/month). No significant difference in bradycardia rates (33% vs. 24%;) and BP (38% vs. 43%) were found between AAN and AN groups, nor were there any difference found in frequency of psychiatric comorbidities (45% vs. 38%).

Author	Population	Analysis	Results	Conclusions
Garber et al. [5]	12 to 24-year-olds with AN (n = 66) and AAN (n = 50)	Randomised controlled trial. AN and AAN groups were statistically compared using t test. Associations between weight history variables (total weight loss, rate of weight loss and duration of weight loss) and indicators of illness severity on admission were examined (HR and serum phosphorus)	Groups did not differ by weight history or admission HR. Independent of admis- sion weight, lower HR ($p = 0.01$) was associated with faster loss; lower serum phosphorus was associated with a greater amount ($p = 0.04$) and longer duration ($p = 0.001$)	Greater rate, duration and amount of weight loss was associated with increased illness severity, regardless of current weight
Meirer et al. [2]	10 to 17-years-olds with AN. 2 cohorts (2004, 2014) 2 groups, premorbid over- weight history ($n = 34$ and $n = 16$ in 2004) and premorbid normal weight history ($n = 72$ and $n = 50$ in 2014)	Single centre retrospective cohort study. Proportion of AN patients with a history of excess weight over a 10-year period examined. Premorbid normal and overweight history groups statistically compared in terms of medical markers at admission (HR, BP, and duration of amen- orrhea) using t-tests	Excess premorbid weight was similar in both cohorts (32% in 2004 versus 29.5% in 2014). The historically overweight subgroup had a lower HR at intake (64.77 versus 69.75, p = 0.03). The total decrease in BMI greater in premorbid overweight group (7 BMI points versus 3.8, p = 0.0001)	Greater rate and amount of weight loss was associated with higher premorbid weight. Screening of AN should not be limited to the underweight and number of AAN cases increasing
Sawyer et al. [15]	13 to 17-year-olds with AAN (n = 42) and AN (n = 118)	Cohort study. Medical markers (HR, BP, temperature, and weight) and psycho- logical markers of adolescents AAN and AN were compared using independent t-tests for continuous variables and X2 for categorical variables	AAN lost more weight (17.6 kg vs 11.0 kg) over a longer period (13.3 vs 10.2 months) than AN. No significant difference in brad- ycardia rates (24% vs. 33%) BP (43% vs. 38%). No difference found in frequency of psychiatric comorbidities (38% vs 45%)	Many adolescents with AAN present with medical instability, despite being healthy weight. Physical and psychological morbid- ity similar across AN and AAN, with similar proportions requiring hospitalisation and rates of AAN increasing
Whitelaw et al. [4]	12 to 19-year-olds with AN (n = 118) and AAN (n = 53)	Retrospective and prospective cohort studies comparing total and recent weight loss with admission weight as pre- dictors of physical (hypophosphatemia, bradycardia, hypothermia, and hypoten- sion) and psychological complications	Greater total weight loss (p = 0.002) and greater recent weight loss (p = 0.006), but not admission weight, were associated with a lower HR. Greater total weight loss (p = 0.003) and greater recent weight loss (p = 0.02) were associated with bradycardia	Number of AAN cases increasing over years. Weight loss was a stronger predictor than degree of underweight of physical compli- cations leading to hospitalisation. Greater attention to weight loss as a measure of starvation is recommended
Whitelaw et al. [38]	12 to 19-year-olds with AN (n = 73) and EDNOS (n = 26)	6-year retrospective cohort study compar- ing HR, BP, and biochemistry of AN and EDNOS adolescents using linear regres- sion for continuous variables, or nonpara- metric rank sum test for non-normal data	Proportion of EDNOS diagnosis increased from 8 to 47% over 6-year period. Hypophosphatemia developed in 41% of AN and in 39% of EDNOS patients. Lowest HR in AN was 45 bpm compared with 47 in EDNOS	Huge increase in the proportion of healthy weight adolescents hospitalised. EDNOS patients experienced a similar degree of medical instability as AN
Hudson et al. [1]	5 to 13-year-olds with AN ($n = 76$), BN ($n = 3$), EDNOS ($n = 83$), OSFED ($n = 46$)	Prospective surveillance study. T-tests and Mann–Whitney U tests were used to compare BMI, weight loss, HR, BP, and temperature between diagnoses	35% of cases had medical instability at presentation (60% bradycardia, 54% hypotension, 34% dehydration, 26% hypothermia), 52% of cases required admission at diagnosis (73% to a pae- diatric ward). 41% of cases had medical instability in the absence of underweight	Anthropological indices alone are poor markers for medical instability, clinical assessment is essential. Doctors providing care for children have central role in both the recognition and management of early onset EDs

Author	Population	Analysis	Results	Conclusions
Ornstein et al. [29]	8 to 22-year-olds with AN (n = 66)	Retrospective chart review. 3 groups (phosphorous < 2.5 mg/dl, < 3.0 mg/dl, and > 3.0 mg/dl) created. Age, weight, % IBW, and phosphorous were compared among groups. Pearson correlation coeffi- cient was used to analyse the relationship between phosphorus and % IBW	Patients who developed moderate hypophosphatemia were significantly more malnourished than those who did not ($p = 0.02$). Phosphorus nadirs were directly proportional to % IBW ($p = 0.01$)	Those patients most severely malnourished are at the greatest risk for developing mod- erate hypophosphatemia
Peebles et al. [3]	8 to 19-year-olds with AN (n = 330), BN (n = 162), and EDNOS (n = 818)	Retrospective chart review. HR, BP, temperature and QTc compared between diagnoses using t-tests and × 2	EDNOS patients had the greatest and fast- est weight lost ($p = 0.001$ and $p = 0.005$) and 28.7% compared with 38.5% AN patients had bradycardia. AN had highest rate of bradycardia ($p = 0.01$) and hypo- tension ($p < 0.001$) and lowest phospho- rous levels ($p < 0.005$)	Medical complications can occur EDNOS and AN. Medical severity of patients with EDNOS intermediate to that of patients with AN
Whitelaw et al. [38]	12 to 18-year-olds with AN (n=29)	Retrospective chart review study. Logistic regression used to identify associations between %BW and phosphorous	%61BW at admission was significantly associated with the subsequent development of hypophosphatemia (p= 0.007)	Significant correlation between %IBW and incidence of hypophosphatemia
N sample size, AN anorexia nervosa, BN bulim weight, QTc electrocardiogram measurement	xia nervosa, <i>BN</i> bulimia nervosa, <i>EDNOS</i> eating c liogram measurement	N sample size, AN anorexia nervosa, BN bulimia nervosa, EDNOS eating disorder not otherwise specified, AAN atypical anorexia nervosa, RCT randomised controlled trial, HR heart rate, BP blood pressure, IBW ideal body weight, QTc electrocardiogram measurement	exia nervosa, <i>RCT</i> randomised controlled trial, <i>HR</i>	heart rate, BP blood pressure, IBW ideal body

Table 3 (continued)

Brennan et al. Journal of Eating Disorders (2023) 11:58

In the retrospective study by Whitelaw et al. (2014), illness severity was compared between adolescents with AN and those presenting with AAN. Results showed that the proportion of adolescents presenting with AAN increased from 8 to 47% over the 6-year period. Hypophosphatemia developed in 41% of AN and in 39% of AAN patients and the lowest HR in AN was 45 bpm compared with 47 bpm in AAN. Authors concluded that adolescents with AAN present with similar physical complications as patients with AN, and that those with greater weight loss should receive close medical monitoring.

Peebles et al. (2010), found that adolescents presenting with AAN, who had lost > 25% of their premorbid body weight were physically compromised (i.e., presented with bradycardia, hypotension, and hypothermia) when compared with those who had not lost significant amounts of weight, independent of degree of underweight. The AAN group had a faster rate of weight loss when compared with the AN group (2.7% vs. 2.4% per month, p=0.005). The AN group, compared with this AAN subset, had the highest rate of bradycardia (38.5% vs. 28.7%, p=0.01) and hypotension (16.4% vs. 5.1%, p<0.001) and highest incidence of hypophosphatemia (8.4% vs. 5.2%, p<0.005). This study concluded that medical severity of patients with AAN was intermediate to that of underweight patients fulfilling all diagnostic criteria for AN.

Weight loss and medical instability

Three studies looked at rate of weight loss in relation to illness severity and physical risk. In the randomised controlled trial by Garber and colleagues 2019, greater speed and magnitude of weight loss were significantly associated with lower HR and lower serum phosphorus, independent of weight. AN and AAN groups did not differ by weight history or HR. Independent of weight, lower HR was associated with faster rate of weight loss (p=0.01); lower serum phosphorus was associated with a greater amount of weight loss (p=0.04) over a longer duration of time (p=0.001). Garber et al. (2019) concluded that weight loss should be considered as a key factor when assessing illness severity, regardless of degree of underweight [5].

In the study by Whitelaw et al. 2018, the relationship between weight loss and illness severity was explored. Greater total weight loss (p=0.002) and greater recent weight loss (p=0.006), but not admission weight, were associated with a lower HR. Greater total weight loss (p=0.003) and greater recent weight loss (p=0.02) were associated with bradycardia, defined as a heart rate lower than 60 bpm. The authors recommended that in adolescents presenting with AN or AAN, these two markers of weight loss were better predictors of medical instability than degree of underweight.

The cohort study conducted by Meirer and colleagues in 2019 investigated the relationship between premorbid overweight and weight history in adolescent AN, as well as the evolution of weight history over a 10-year period. Excess premorbid weight was similar in both the historically overweight and historically healthy weight cohorts (32% in 2004 versus 29.5% in 2014). The historically overweight subgroup had a lower HR at intake (64.77 versus 69.75, p = 0.03). The total decrease in BMI was greater in premorbid overweight group (seven BMI points versus 3.8, p = 0.0001). The authors recommended that with the number of AAN cases increasing, screening should not be limited to just the underweight.

Degree of underweight and hypophosphatemia

The remaining two articles explored the relationship between degree of underweight and risk of refeeding syndrome complications, namely hypophosphatemia. Degree of underweight was expressed as %IBW rather than %mBMI in both studies. Ornstein and colleagues in 2003 found that hypophosphatemia was directly proportional to degree of underweight. Patients who developed moderate hypophosphatemia were significantly more malnourished than those who did not (p=0.02). Phosphorus nadirs were directly proportional to degree of underweight (p=0.01). Incidence of hypophosphatemia was greatest in adolescents that were less than 70% IBW. Extreme malnutrition and lower weight were concluded to put patients at greatest risk of hypophosphatemia.

The study by Whitelaw et al. (2010) had similar findings. Lower body mass at admission was significantly associated with development of hypophosphatemia (p=0.007). Conclusions from this study, similar to that of Ornstein 2003, found that malnourished patients less than 70% IBW were at greatest risk of hypophosphatemia and that greater caution should be used in their management.

Meta-analysis

Of the nine studies included in the narrative review, three of these contained data that could be extracted and statistically analysed. Authors of studies with missing data were contacted but failed to respond with required supplementary data for analysis. Studies included in metaanalysis were Garber et al. (2019), Sawyer et al. (2016), and Whitelaw et al. (2014). A meta-analysis was conducted where three or more studies contained data related to markers of medical instability that could be compared. Forest plots were generated and results of these are displayed in Figs. 2, 3, 4, 5.



Fig. 2 Forest plot showing association between degree of underweight and lower heart rate in adolescents with eating disorders



Fig. 3 Forest plot showing association between degree of underweight and lower blood pressure in adolescents with eating disorders

	Unde	rweig	ht	Non-ur	nderwe	ight		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% CI
Garber 2019	36.6	0.1	66	36.6	0.1	50	53.3%	0.00 [-0.04, 0.04]	•
Saywer 2016	36.4	0.7	118	36.8	0.7	42	43.8%	-0.40 [-0.65, -0.15]	•
Whitelaw 2014	35.9	0.4	73	34.8	5.7	26	2.9%	1.10 [-1.09, 3.29]	1 7
Total (95% CI)			257			118	100.0%	-0.14 [-0.53, 0.24]	
Heterogeneity: Tau ² = Test for overall effect:				r= 2 (P =	0.004);	I ² = 82	%		-100 -50 0 50 100 Favours underweight Favours non-underweight

Fig. 4 Forest plot showing association between degree of underweight and lower temperature in adolescents with eating disorders

	Unde	rweig	ht	Non-ur	nderwe	ight		Mean Difference		Mean Dif	ference	
Study or Subgroup	Mean	SD	Total	Mean	\$D	Total	Weight	IV, Random, 95% CI		IV, Randor	n, 95% Cl	
Garber 2019	1.6	1.4	66	1.6	1.3	50	33.1%	0.00 [-0.49, 0.49]		•		
Saywer 2016	1.1	0.9	118	1.3	1.2	42	39.6%	-0.20 [-0.60, 0.20]			l i i i i i i i i i i i i i i i i i i i	
Whitelaw 2014	1.4	1.4	73	2.2	1.3	26	27.4%	-0.80 [-1.39, -0.21]				
Total (95% CI)			257			118	100.0%	-0.30 [-0.71, 0.12]				
Heterogeneity: Tau ² = Test for overall effect:				= 2 (P = 0).12); l²	= 54%			-100	-50 0 Favours underweight	-	100

Fig. 5 Forest plot showing association between degree of underweight and greater rate of weight loss in adolescents with eating disorders

Table 4 Characteristics of	f underweight (AN) and non-un	derweight (AAN) adolescents	compared by meta-analysis

	n	HR (bpm)	Temperature (°C)	BP (mmHg)	mBMI	Phosphorous (mmol/L)
AN(n = 257)						
Garber et al. [5]	66	42 (7)	36.6 (0.1)	92 (7.6)	76.5% (5.9)	1.2 (0.2)
Whitelaw et al. [38]	73	45.1 (9.9)	35.9 (0.4)	84 (10)	74.5% (7.1)	1.2 (0.1)
Sawyer et al. [15]	118	59.3 (15)	36.4 (0.7)	99.1 (11.3)	79% (7.1)	n/a
AAN (n = 118)						
Garber et al. [5]	50	40.3 (4.7)	36.6 (0.1)	96.4 (8.1)	95.2% (9)	1.3 (0.2)
Whitelaw et al. [38]	26	47.1 (13.7)	34.8 (0.7)	91 (7)	93% (5.3)	1.2 (0.1)
Sawyer et al. [15]	42	59.8 (11)	36.8 (0.7)	106.2 (10.3)	106% (13.4)	n/a

In each of the four analyses, two independent groups (underweight AN and non-underweight AAN) were created and compared (Table 4). The underweight group included adolescents diagnosed with AN (n=257), fulfilling full criteria for the disorder as specified by the DSM-5 including the presence of underweight. Median BMI ranged from 74.5 to 79%. The non-underweight group (n=118) included adolescents who fulfilled partial criteria for AN but were not underweight, thus diagnosed with OSFED or EDNOS, and described as the AAN group in this review. Median BMI ranged from 93 to 106%.

Groups were comparable in terms of age and gender and differed by their percentage median BMI. Both groups included male and female adolescents, presenting with symptoms of AN. Heart rate, BP, temperature, and rate of weight loss at the time of initial assessment were statistically compared between groups. Serum phosphorus could not be compared as only two studies contained data related to this marker. Thus, the relationship between hypophosphatemia and degree of underweight could not be fully investigated.

Results identified a statistically significant relationship between degree of underweight and blood pressure. Underweight adolescents were more likely to have lower BP than those that were not underweight (p < 0.00001). There was no statistically significant association found between degree of underweight and any other markers investigated in the analysis, including HR, temperature, and rate of weight loss (p=0.31, p=0.46 and p=0.16, respectively).

Discussion

Overview of results

This review aimed to determine firstly, the relationship between weight and medical instability in adolescents with typical and atypical anorexia nervosa, and secondly, if underweight adolescents with AN are at greater risk of medical instability than non-underweight adolescents with AAN.

Findings from the narrative review identified that medical instability occurs in a large proportion of adolescents presenting with AN and AAN, across a range of body weights. Results indicated that the speed and magnitude of weight loss perpetuate medical instability in both underweight and non-underweight groups. Low heart rate and blood pressure were the most likely symptoms of malnutrition to occur across both groups.

Only a small number of studies (n=3) reported results that could be included in statistical analysis. Results from meta-analysis identified underweight adolescents with AN as having significantly lower BP than nonunderweight adolescents presenting with AAN. All other markers of medical instability were similar across groups, with no significant differences identified.

Medical instability in AN and AAN presentations

Evidence has shown that prevalence of AAN has increased over the past decade [4]. Despite presenting as clinically non-underweight, hospital admissions are often required in this group, with research showing similar rates to their underweight counterparts [2, 15, 39]. Considering this growing number of medically unstable, but non-underweight adolescents, comprehensive medical assessment of all adolescents presenting with restrictive eating disorders is recommended [1, 2].

Results from this study indicated that medical instability and complications related to starvation were not limited to underweight adolescents. Healthy weight adolescents were found to have a similar degree of medical instability as those classed as underweight, although the lower weight group did have significantly lower blood pressures. Bradycardia and hypotension were the most common presenting symptoms, and rates were similar across underweight AN and non-underweight AAN groups. Rates of bradycardia ranged from 29 to 60% and rates of hypotension ranged from 40 to 54%, across all diagnoses. Hospitalisation was also common across groups, with~40% of non-underweight adolescents in studies reviewed requiring a medical admission. These results correlate with findings of other recent reviews that compare AN and AAN [11, 34].

Statistical analyses found that underweight adolescents with AN had significantly lower BP than non-underweight adolescents with AAN. However, these results were limited by the few studies containing sufficient data that could be analysed (three out of nine studies). Additionally, groups compared by meta-analysis did not differ significantly in rates of weight loss preceding assessment. Therefore, results should be interpreted cautiously, with underweight adolescents being at significantly higher risk of having lower BP than those that are not underweight, in the absence of significant differences in rates of weight loss. Results may differ in non-underweight adolescents that have lost significant amounts of weight. The severity of certain medical markers, such as BP, is likely worsened in adolescents who present with extremes of low weight in addition to rapid weight loss.

Medical instability and weight loss

In adult AN, weight suppression (difference between weight at assessment and premorbid weight) has emerged as a marker of illness severity, independent of degree of underweight [5]. Weight suppression, in individuals presenting at normal weights, has been associated with medical instability (e.g., bradycardia) [4]. In the RCT conducted by Garber and colleagues in 2019, weight loss was identified as a key marker of malnutrition and determinant of hospitalisation, independent of other factors [5].

In our review, findings indicated that greater magnitude and speed of weight loss predicts physical complications requiring hospitalisation, independent of degree of underweight. In addition, adolescents of a higher premorbid weight were found to have greater weight loss than those of a lower weight [4, 5]. Results from the narrative part of this review support weight loss as a predictor of medical complications, in addition to degree of underweight. Adolescents presenting with greater total, duration, and rate of weight loss in addition to low weight are at greatest risk of bradycardia and hypotension. However, due to the lack of statistically significant difference in weight loss rates between underweight and nonunderweight groups, meta-analysis could not be used to compare physical risk by weight loss rate.

Hypophosphatemia and weight

All reviewed studies were in agreement that a positive relationship between the degree of malnutrition and hypophosphatemia on admission exists [40]. The degree of malnutrition could be used as an indicator for pending hypophosphatemia but is not a definitive risk factor, with both underweight and non-underweight adolescents with AN and AAN being at risk [31].

Hypophosphatemia was identified across both underweight and non-underweight groups. However, results from the narrative review identified adolescents < 70% mBMI as being at significantly greater risk of developing this symptom. Considering the findings of this review, paired with results from recent research studies, adolescents < 70%mBMI may be considered at greatest risk of developing hypophosphatemia during nutritional rehabilitation [29, 32]. Other factors, such as rapid weight loss, further increase this risk [5].

Strengths

The key strength of this review stands in it being one of the few systematic reviews of this nature, looking specifically at evidence relating to weight and risk of medical instability in adolescents with typical and atypical anorexia nervosa. Although there have been multiple review papers that look at anorexia nervosa and physical health, none provide a clear analysis of the objective significance of weight parameters in assessing risk of physical deterioration.

This review brings together the limited evidence that is available on this topic. It highlights the need for further research studies investigating the factors associated with physical risk in AN and AAN, across underweight and non-underweight adolescents.

Limitations

The results of this review are limited by the small number of studies that met inclusion criteria and that could be analysed. Many studies included in the narrative review lacked comparable data, meaning they were excluded from statistical analysis, limiting the power of the findings of the meta-analysis. Ethnic variation could not be considered in meta-analysis. This limits the accuracy of results due to cut-offs for thinness varying across ethnicities. Similarly, the high degree of heterogeneity across the sample limits the power of the results from this statistical analysis. Different methods of expressing degree of underweight were used across studies included in the narrative review, limiting the comparability of study results. However, studies included in the statistical analysis were consistent in their use of %mBMI to express degree of underweight, which minimises the effect this factor had on these results.

There is a growing body of evidence on adolescent eating disorders. However, high quality intervention studies are hugely lacking. This review identified many studies for inclusion, however, there was a lack comparable data that could be quantitively analysed.

Conclusions

Current National guidance specifies the range of risks associated with starvation and stratifies this risk for each marker [24]. Greatest risk is attributed to those < 70% mBMI, as well as those presenting with greatest weight loss. This review draws together currently available evidence and indicates that greatest risk of medical instability occurs in adolescents who are most severely malnourished. This may be evidenced by their extremes of low body weight, their rapid weight loss, or a combination of these two factors.

Future work should focus on studies that can be compared to current evidence, to allow a more thorough analysis of data relating to weight parameters and their significance in typical and atypical adolescent anorexia nervosa. Further research is essential in this area, particularly in non-underweight adolescents with AAN to help clinicians understand medical risks arising in the absence of underweight.

Abbreviations

ED	Eating disorder
AN	Anorexia nervosa
DSM	Diagnostic and statistical manual
BMI	Body mass index
%mBMI	Percentage median body mass index

Kilogram
Metres squared
Weight for height
Ideal body weight
Eating disorder not otherwise specified
Atypical anorexia nervosa
Otherwise specified feeding or eating disorder
Managing emergencies in eating disorders
Beats per minute
Electrocardiogram
Refeeding syndrome
Confidence interval
Millimetres of mercury
Randomised controlled trial

Acknowledgements

I would like to acknowledge Shawn McLaren for his help with the statistics used in this review.

Author contributions

CB and DB conceptualised the study performed the literature search and review. DB is the PhD supervisor of CB. All authors read and reviewed the final manuscript, figures and tables.

Funding

Not applicable

Availability of data and materials

All data generated or analysed during this study are included in this published article.

Declarations

Ethics approval and consent to participate Not applicable

Consent for publication

Not applicable

Competing interests

The authors declare that they have no competing interests.

Author details

¹South London and Maudsley NHS Trust, Maudsley Hospital, De Crespigny Park, London SE5 8AZ, UK. ²London Metropolitan University, 166-220 Holloway Road, London N7 8DB, UK. ³ East London NHS Foundation Trust, Emmanuel Miller Centre, 11 Gill Street, London E14 8HQ, UK. ⁴Institute of Psychiatry, Psychology & Neuroscience, Kings College London, De Crespigny Park, London SE5 8AB, UK. ⁵University College London, London, UK.

Received: 26 January 2023 Accepted: 27 March 2023 Published online: 06 April 2023

References

- Hudson LD, Nicholls DE, Lynn RM, Viner RM. Medical instability and growth of children and adolescents with early onset eating disorders. Arch Dis Child. 2012;97(9):779–84.
- Meierer K, Hudon A, Sznajder M, Leduc MF, Taddeo D, Jamoulle O, et al. Anorexia nervosa in adolescents: evolution of weight history and impact of excess premorbid weight. Eur J Pediatr. 2018. https://doi.org/10.1007/ s00431-018-3275-y.
- Peebles R, Hardy KK, Wilson JL, Lock JD. Are diagnostic criteria for eating disorders markers of medical severity? Pediatrics. 2010. https://doi.org/10. 1542/peds.2008-1777.
- Whitelaw M, Lee KJ, Gilbertson H, Sawyer SM. Predictors of complications in anorexia nervosa and atypical anorexia nervosa: Degree of underweight or extent and recency of weight loss? J Adolesc Heal. 2018;63(6):717–23.

- Garber AK, Cheng J, Accurso EC, Adams SH, Buckelew SM, Kapphahn CJ, et al. Weight loss and illness severity in adolescents with atypical anorexia nervosa. Pediatrics. 2019. https://doi.org/10.1542/peds.2019-2339.
- Buchman S, Attia E, Dawson L, Steinglass JE. Steps of care for adolescents with anorexia nervosa—a Delphi study. Int J Eat Disord. 2019;52(7):777–85.
- Cole TJ, Flegal KM, Nicholls D, Jackson AA. Body mass index cut offs to define thinness in children and adolescents: international survey. Br Med J. 2007;335(7612):194–7.
- Kang K, Absher R, Farrington E, Ackley R, So TY. Evaluation of different methods used to calculate ideal body weight in the pediatric population. J Pediatr Pharmacol Ther. 2019;24(5):421–30.
- American Psychiatric Association. Diagnostic and statistical manual of mental disorders. 4th ed. Washington, D.C.: American Psychiatric Association; 2000.
- Thomas JJ, Vartanian LR, Brownell KD. The relationship between eating disorder not otherwise specified (EDNOS) and officially recognized eating disorders: meta-analysis and implications for DSM. Am Psychol Assoc. 2009;135(3):407–33.
- Moskowitz L, Weiselberg E. Anorexia nervosa/atypical anorexia nervosa. Curr Probl Pediatr Adolesc Health Care. 2017;47(4):70–84.
- 12. The American Psychiatric Association. The Diagnostic and statistical manual of mental disorders. 5th ed. Washington, D.C.: American Psychiatric Association; 2013. p. 307.
- Fisher M, Gonzalez M, Malizio J. Eating disorders in adolescents: How does the DSM-5 change the diagnosis? Int J Adolesc Med Health. 2015;27(4):437–41.
- 14. Hay P. Current approach to eating disorders: a clinical update. Intern Med J. 2020;50(1):24–9.
- Sawyer SM, Whitelaw M, Le Grange D, Yeo M, Hughes EK. Physical and psychological morbidity in adolescents with atypical anorexia nervosa. Pediatrics. 2016. https://doi.org/10.1542/peds.2015-4080.
- Smink FRE, Van Hoeken D, Hoek HW. Epidemiology of eating disorders: incidence, prevalence and mortality rates. Curr Psychiatry Rep. 2012;14(4):406–14.
- Frank GKW, DeGuzman MC, Shott ME. Motivation to eat and not to eat-the psycho-biological conflict in anorexia nervosa. Physiol Behav. 2019;206(October 2018):185–90.
- Golden NH, Katzman DK, Sawyer SM, Ornstein RM, Rome ES, Garber AK, et al. Update on the medical management of eating disorders in adolescents. J Adolesc Heal. 2015;56(4):370–5.
- Katzman DK. Medical complications in adolescents with anorexia nervosa: a review of the literature. Int J Eat Disord. 2005. https://doi. org/10.1002/eat.20118.
- 20. Hudson LD, Chapman S. Paediatric medical care for children and young people with eating disorders: achievements and where to next. Clin Child Psychol Psychiatry. 2020;25(3):716–20.
- Neale J, Hudson LD. Anorexia nervosa in adolescents. Br J Hosp Med. 2020;81(6):1–8.
- 22. Chidiac CW. An update on the medical consequences of anorexia nervosa. Curr Opin Pediatr. 2019;31(4):448–53.
- Misra M, Aggarwal A, Miller K, Almazan C, Worley M, Soyka L, et al. Effects of anorexia nervosa on clinical, hematologic, biochemical, and bone density parameters in community-dwelling adolescent girls. Pediatrics. 2004;116(6):1574–83.
- Royal College of Psychiatrists. CR 233: medical emergencies in eating disorders (MEED) Guidance on recognition and management. 2022;(May):185. Available from: https://www.rcpsych.ac.uk/docs/defau lt-source/improving-care/better-mh-policy/college-reports/collegereport-cr233-medical-emergencies-in-eating-disorders-(meed)-guida nce.pdf?sfvrsn=2d327483_50
- DiVasta AD, Walls CE, Feldman HA, Quach AE, Woods ER, Gordon CM, et al. Malnutrition and hemodynamic status in adolescents hospitalized for anorexia nervosa. Arch Pediatr Adolesc Med. 2010;164(8):706–13.
- 26. Assalone C, Leonardi L, Franceschi R, Fumanelli J, Maines E, Marini M, et al. Determinants of severe bradycardia in adolescents hospitalized for anorexia nervosa. Pediatr Int. 2022;64(1): e14967.
- Morris R, Prasad A, Asaro J, Guzman M, Sanders L, Hauck A, et al. Markers of Cardiovascular Dysfunction in Adolescents With Anorexia Nervosa. Glob Pediatr Heal. 2017;4:0–5.

- O'Connor G, Nicholls D. Refeeding hypophosphatemia in adolescents with anorexia nervosa: a systematic review. Nutr Clin Pract. 2013;28(3):358–64.
- Ornstein RM, Golden NH, Jacobson MS, Shenker IR. Hypophosphatemia during nutritional rehabilitation in anorexia nervosa: Implications for refeeding and monitoring. J Adolesc Heal. 2003;32(1):83–8.
- Davies JE, Cockfield A, Brown A, Corr J, Smith D, Munro C. The medical risks of severe anorexia nervosa during initial re-feeding and medical stabilisation. Clin Nutr ESPEN. 2017;1(17):92–9.
- O'Connor G, Nicholls D, Hudson L, Singhal A. Refeeding low weight hospitalized adolescents with anorexia nervosa: a multicenter randomized controlled trial. Nutr Clin Pract. 2016;31(5):681–9.
- Garber A, Cheng J, Accurso E. Short-term outcomes of the study of refeeding to optimize inpatient gains for patients with anorexia nervosa: a multicenter randomized clinical trial. JAMA Pediatr. 2020. https://doi.org/10.1001/jamapediatrics.2020.3359.
- Withnell SJ, Kinnear A, Masson P, Bodell LP. How different are threshold and other specified feeding and eating disorders? Comparing severity and treatment outcome. Front Psychol. 2022;13(February):1–6.
- Freizinger M, Recto M, Jhe G, Lin J. Atypical anorexia in youth: cautiously bridging the treatment gap. Children. 2022;9(6):1–13.
- Strand M, Zvrskovec J, Hübel C, Peat CM, Bulik CM, Birgegård A. Identifying research priorities for the study of atypical anorexia nervosa: a Delphi study. Int J Eat Disord. 2020;53(10):1729–38.
- Walsh BT, Hagan KE, Lockwood C. A systematic review comparing atypical anorexia nervosa and anorexia nervosa. Int J Eat Disord. 2022. https:// doi.org/10.1002/eat.23856.
- Fletcher J. What is heterogeneity and is it important? Br Med J. 2007;330:94–6.
- Whitelaw M, Gilbertson H, Lam PY, Sawyer SM. Does aggressive refeeding in hospitalized adolescents with anorexia nervosa result in increased hypophosphatemia? J Adolesc Heal. 2010;46(6):577–82. https://doi.org/ 10.1016/j.jadohealth.2009.11.207.
- Whitelaw M, Gilbertson H, Lee KJ, Sawyer SM. Restrictive eating disorders among adolescent inpatients. Pediatrics. 2014;134(3):e758–64.
- Dalenbrook S, Naab S, Garber AK, Correll CU, Voderholzer U, Haas V. Outcomes of a standardized, high-caloric, inpatient re-alimentation treatment protocol in 120 severely malnourished adolescents with anorexia nervosa. J Clin Med. 2022. https://doi.org/10.3390/jcm11092585.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

